
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Tue May  5 16:28:31 GMT 1998
Year: 98  Doy: 125
Observer: yasukawa
WEATHER COMMENT: Tue May  5 16:29:56 GMT 1998
Cool, in continued thick cirrus overcast.  Yesterday’s vacation day
probably resulted in no loss of a chance for good data due to overcast.
Light south breeze.
LOW-L COMMENT: Tue May  5 16:32:29 GMT 1998
L00512 found popped out of temporary drive 0.  5 files of data already
written to temporary drive 1.  Drive failure must have occurred 
yesterday, pointing to a flaky exabyte drive as the short tapes occur
in the top drive regardless which interface designation is writing to it.
Cleaned drive and loaded L00514.  Data currently writing to L00513 in 
temporary drive 1.
MKIII COMMENT: Tue May  5 16:36:39 GMT 1998

*** Recycling H01815, H01816, and H01817 ***

COMMENT: Tue May  5 17:14:54 GMT 1998
Going down to GONG to do the weekly and monthly preventive maintenance.
WEATHER COMMENT: Tue May  5 20:08:18 GMT 1998
Cirrus overcast continues.  Shutting down for the day.
No observations.
Tue May  5 20:10:44 GMT 1998 
        MkIII   
 


